\begin{thebibliography}{}
\expandafter\ifx\csname natexlab\endcsname\relax\def\natexlab#1{#1}\fi
\providecommand{\url}[1]{\href{#1}{#1}}
\providecommand{\dodoi}[1]{doi:~\href{http://doi.org/#1}{\nolinkurl{#1}}}
\providecommand{\doeprint}[1]{\href{http://ascl.net/#1}{\nolinkurl{http://ascl.net/#1}}}
\providecommand{\doarXiv}[1]{\href{https://arxiv.org/abs/#1}{\nolinkurl{https://arxiv.org/abs/#1}}}

\bibitem[{{Amard} {et~al.}(2019){Amard}, {Palacios}, {Charbonnel}, {Gallet},
  {Georgy}, {Lagarde}, \& {Siess}}]{Amard2019}
{Amard}, L., {Palacios}, A., {Charbonnel}, C., {et~al.} 2019, \aap, 631, A77,
  \dodoi{10.1051/0004-6361/201935160}

\bibitem[{{Bahcall} {et~al.}(2001){Bahcall}, {Pinsonneault}, \&
  {Basu}}]{Bahcall2001}
{Bahcall}, J.~N., {Pinsonneault}, M.~H., \& {Basu}, S. 2001, \apj, 555, 990,
  \dodoi{10.1086/321493}

\bibitem[{{Bahcall} {et~al.}(2005){Bahcall}, {Serenelli}, \&
  {Basu}}]{Bahcall2005}
{Bahcall}, J.~N., {Serenelli}, A.~M., \& {Basu}, S. 2005, \apjl, 621, L85,
  \dodoi{10.1086/428929}

\bibitem[{{Baraffe} \& {Chabrier}(2018)}]{Baraffe2018}
{Baraffe}, I., \& {Chabrier}, G. 2018, \aap, 619, A177,
  \dodoi{10.1051/0004-6361/201834062}

\bibitem[{{Baraffe} {et~al.}(1997){Baraffe}, {Chabrier}, {Allard}, \&
  {Hauschildt}}]{Baraffe1997}
{Baraffe}, I., {Chabrier}, G., {Allard}, F., \& {Hauschildt}, P.~H. 1997, \aap,
  327, 1054.
\newblock \doarXiv{astro-ph/9704144}

\bibitem[{Boudreaux(2022)}]{Boudreaux22}
Boudreaux, T. 2022, tboudreaux/pytopsscrape: pyTOPSScrape v1.0, v1.0,  Zenodo,
  \dodoi{10.5281/zenodo.7094198}

\bibitem[{{Chabrier} \& {Baraffe}(1997)}]{Chabrier1997}
{Chabrier}, G., \& {Baraffe}, I. 1997, \aap, 327, 1039.
\newblock \doarXiv{astro-ph/9704118}

\bibitem[{Chandra \& Varanasi(2015)}]{chandra2015python}
Chandra, R.~V., \& Varanasi, B.~S. 2015, Python requests essentials (Packt
  Publishing Ltd)

\bibitem[{{Choi} {et~al.}(2016){Choi}, {Dotter}, {Conroy}, {Cantiello},
  {Paxton}, \& {Johnson}}]{Choi2016}
{Choi}, J., {Dotter}, A., {Conroy}, C., {et~al.} 2016, \apj, 823, 102,
  \dodoi{10.3847/0004-637X/823/2/102}

\bibitem[{{Colgan} {et~al.}(2016){Colgan}, {Kilcrease}, {Magee}, {Sherrill},
  {Abdallah}, {Hakel}, {Fontes}, {Guzik}, \& {Mussack}}]{Colgan2016}
{Colgan}, J., {Kilcrease}, D.~P., {Magee}, N.~H., {et~al.} 2016, in APS Meeting
  Abstracts, Vol. 2016, APS Division of Atomic, Molecular and Optical Physics
  Meeting Abstracts, D1.008

\bibitem[{{Creevey} {et~al.}(2022){Creevey}, {Sordo}, {Pailler}, {Fr{\'e}mat},
  {Heiter}, {Th{\'e}venin}, {Andrae}, {Fouesneau}, {Lobel}, {Bailer-Jones},
  {Garabato}, {Bellas-Velidis}, {Brugaletta}, {Lorca}, {Ordenovic}, {Palicio},
  {Sarro}, {Delchambre}, {Drimmel}, {Rybizki}, {Torralba Elipe}, {Korn},
  {Recio-Blanco}, {Schultheis}, {De Angeli}, {Montegriffo}, {Abreu Aramburu},
  {Accart}, {{\'A}lvarez}, {Bakker}, {Brouillet}, {Burlacu}, {Carballo},
  {Casamiquela}, {Chiavassa}, {Contursi}, {Cooper}, {Dafonte}, {Dapergolas},
  {de Laverny}, {Dharmawardena}, {Edvardsson}, {Le Fustec},
  {Garc{\'\i}a-Lario}, {Garc{\'\i}a-Torres}, {Gomez},
  {Gonz{\'a}lez-Santamar{\'\i}a}, {Hatzidimitriou}, {Jean-Antoine Piccolo},
  {Kontizas}, {Kordopatis}, {Lanzafame}, {Lebreton}, {Licata}, {Lindstr{\o}m},
  {Livanou}, {Magdaleno Romeo}, {Manteiga}, {Marocco}, {Marshall}, {Mary},
  {Nicolas}, {Pallas-Quintela}, {Panem}, {Pichon}, {Poggio}, {Riclet}, {Robin},
  {Santove{\~n}a}, {Silvelo}, {Slezak}, {Smart}, {Soubiran}, {S{\"u}veges},
  {Ulla}, {Utrilla}, {Vallenari}, {Zhao}, {Zorec}, {Barrado}, {Bijaoui},
  {Bouret}, {Blomme}, {Brott}, {Cassisi}, {Kochukhov}, {Martayan}, {Shulyak},
  \& {Silvester}}]{Creevey2022}
{Creevey}, O.~L., {Sordo}, R., {Pailler}, F., {et~al.} 2022, arXiv e-prints,
  arXiv:2206.05864.
\newblock \doarXiv{2206.05864}

\bibitem[{Dierckx(1981)}]{Dierckx1981}
Dierckx, P. 1981, IMA Journal of Numerical Analysis, 1, 267,
  \dodoi{10.1093/imanum/1.3.267}

\bibitem[{Dotter {et~al.}(2008)Dotter, Chaboyer, Jevremovi{\'c}, Kostov, Baron,
  \& Ferguson}]{Dotter2008}
Dotter, A., Chaboyer, B., Jevremovi{\'c}, D., {et~al.} 2008, The Astrophysical
  Journal Supplement Series, 178, 89

\bibitem[{{Feiden} {et~al.}(2021){Feiden}, {Skidmore}, \& {Jao}}]{Feiden2021}
{Feiden}, G.~A., {Skidmore}, K., \& {Jao}, W.-C. 2021, \apj, 907, 53,
  \dodoi{10.3847/1538-4357/abcc03}

\bibitem[{{Ferguson} {et~al.}(2005){Ferguson}, {Alexander}, {Allard}, {Barman},
  {Bodnarik}, {Hauschildt}, {Heffner-Wong}, \& {Tamanai}}]{Ferguson2005}
{Ferguson}, J.~W., {Alexander}, D.~R., {Allard}, F., {et~al.} 2005, \apj, 623,
  585, \dodoi{10.1086/428642}

\bibitem[{{Fontes} {et~al.}(2015){Fontes}, {Zhang}, {Abdallah}, {Clark},
  {Kilcrease}, {Colgan}, {Cunningham}, {Hakel}, {Magee}, \&
  {Sherrill}}]{Fontes2016}
{Fontes}, C.~J., {Zhang}, H.~L., {Abdallah}, J., J., {et~al.} 2015, Journal of
  Physics B Atomic Molecular Physics, 48, 144014,
  \dodoi{10.1088/0953-4075/48/14/144014}

\bibitem[{{Gaia Collaboration} {et~al.}(2021){Gaia Collaboration}, {Smart},
  {Sarro}, {Rybizki}, {Reyl{\'e}}, {Robin}, {Hambly}, {Abbas}, {Barstow}, {de
  Bruijne}, \& et~al.}]{GaiaCollaboration2021}
{Gaia Collaboration}, {Smart}, R.~L., {Sarro}, L.~M., {et~al.} 2021, \aap, 649,
  A6, \dodoi{10.1051/0004-6361/202039498}

\bibitem[{{Grevesse} \& {Sauval}(1998)}]{Grevesse1998}
{Grevesse}, N., \& {Sauval}, A.~J. 1998, \ssr, 85, 161,
  \dodoi{10.1023/A:1005161325181}

\bibitem[{{Hakel} {et~al.}(2006){Hakel}, {Sherrill}, {Mazevet}, {Abdallah},
  {Colgan}, {Kilcrease}, {Magee}, {Fontes}, \& {Zhang}}]{Hakel2006}
{Hakel}, P., {Sherrill}, M.~E., {Mazevet}, S., {et~al.} 2006, \jqsrt, 99, 265,
  \dodoi{10.1016/j.jqsrt.2005.04.007}

\bibitem[{{Iglesias} \& {Rogers}(1996)}]{Iglesias1996}
{Iglesias}, C.~A., \& {Rogers}, F.~J. 1996, \apj, 464, 943,
  \dodoi{10.1086/177381}

\bibitem[{{Irwin}(2012)}]{Irwin2012}
{Irwin}, A.~W. 2012, {FreeEOS: Equation of State for stellar interiors
  calculations}, Astrophysics Source Code Library, record ascl:1211.002.
\newblock \doeprint{1211.002}

\bibitem[{{Jao} \& {Feiden}(2020)}]{Jao2020}
{Jao}, W.-C., \& {Feiden}, G.~A. 2020, \aj, 160, 102,
  \dodoi{10.3847/1538-3881/aba192}

\bibitem[{{Jao} \& {Feiden}(2021)}]{Jao2021}
---. 2021, Research Notes of the American Astronomical Society, 5, 124,
  \dodoi{10.3847/2515-5172/ac053a}

\bibitem[{{Jao} {et~al.}(2018){Jao}, {Henry}, {Gies}, \& {Hambly}}]{Jao2018}
{Jao}, W.-C., {Henry}, T.~J., {Gies}, D.~R., \& {Hambly}, N.~C. 2018, \apjl,
  861, L11, \dodoi{10.3847/2041-8213/aacdf6}

\bibitem[{{Jermyn} {et~al.}(2022){Jermyn}, {Bauer}, {Schwab}, {Farmer}, {Ball},
  {Bellinger}, {Dotter}, {Joyce}, {Marchant}, {Mombarg}, {Wolf}, {Wong},
  {Cinquegrana}, {Farrell}, {Smolec}, {Thoul}, {Cantiello}, {Herwig}, {Toloza},
  {Bildsten}, {Townsend}, \& {Timmes}}]{Jermyn2022}
{Jermyn}, A.~S., {Bauer}, E.~B., {Schwab}, J., {et~al.} 2022, arXiv e-prints,
  arXiv:2208.03651.
\newblock \doarXiv{2208.03651}

\bibitem[{{Joyce} \& {Chaboyer}(2018)}]{Joyce2018}
{Joyce}, M., \& {Chaboyer}, B. 2018, \apj, 864, 99,
  \dodoi{10.3847/1538-4357/aad464}

\bibitem[{{MacDonald} \& {Gizis}(2018)}]{MacDonald2018}
{MacDonald}, J., \& {Gizis}, J. 2018, \mnras, 480, 1711,
  \dodoi{10.1093/mnras/sty1888}

\bibitem[{{Magee} {et~al.}(2004){Magee}, {Abdallah}, {Colgan}, {Hakel},
  {Kilcrease}, {Mazevet}, {Sherrill}, {Fontes}, \& {Zhang}}]{Magee2004}
{Magee}, N.~H., {Abdallah}, J., {Colgan}, J., {et~al.} 2004, in American
  Institute of Physics Conference Series, Vol. 730, Atomic Processes in
  Plasmas: 14th APS Topical Conference on Atomic Processes in Plasmas, ed.
  J.~S. {Cohen}, D.~P. {Kilcrease}, \& S.~{Mazavet}, 168--179,
  \dodoi{10.1063/1.1824868}

\bibitem[{{Mansfield} \& {Kroupa}(2021)}]{Mansfield2021}
{Mansfield}, S., \& {Kroupa}, P. 2021, \aap, 650, A184,
  \dodoi{10.1051/0004-6361/202140536}

\bibitem[{{Nutzman} \& {Charbonneau}(2008)}]{Nut08}
{Nutzman}, P., \& {Charbonneau}, D. 2008, \pasp, 120, 317,
  \dodoi{10.1086/533420}

\bibitem[{Richardson(2007)}]{richardson2007beautiful}
Richardson, L. 2007, April

\bibitem[{{Rodr{\'\i}guez-L{\'o}pez}(2019)}]{Rodriguez-Lopez2019}
{Rodr{\'\i}guez-L{\'o}pez}, C. 2019, Frontiers in Astronomy and Space Sciences,
  6, 76, \dodoi{10.3389/fspas.2019.00076}

\bibitem[{{Seaton} {et~al.}(1994){Seaton}, {Yan}, {Mihalas}, \&
  {Pradhan}}]{Seaton1994}
{Seaton}, M.~J., {Yan}, Y., {Mihalas}, D., \& {Pradhan}, A.~K. 1994, \mnras,
  266, 805, \dodoi{10.1093/mnras/266.4.805}

\bibitem[{{Skrutskie} {et~al.}(2006){Skrutskie}, {Cutri}, {Stiening},
  {Weinberg}, {Schneider}, {Carpenter}, {Beichman}, {Capps}, {Chester},
  {Elias}, {Huchra}, {Liebert}, {Lonsdale}, {Monet}, {Price}, {Seitzer},
  {Jarrett}, {Kirkpatrick}, {Gizis}, {Howard}, {Evans}, {Fowler}, {Fullmer},
  {Hurt}, {Light}, {Kopan}, {Marsh}, {McCallon}, {Tam}, {Van Dyk}, \&
  {Wheelock}}]{Skrutskie2006}
{Skrutskie}, M.~F., {Cutri}, R.~M., {Stiening}, R., {et~al.} 2006, \aj, 131,
  1163, \dodoi{10.1086/498708}

\bibitem[{Sollima(2019)}]{Sollima2019}
Sollima, A. 2019, Monthly Notices of the Royal Astronomical Society, 489, 2377,
  \dodoi{10.1093/mnras/stz2093}

\bibitem[{{Spada} {et~al.}(2021){Spada}, {Demarque}, \& {Kupka}}]{Spada2021}
{Spada}, F., {Demarque}, P., \& {Kupka}, F. 2021, \mnras, 504, 3128,
  \dodoi{10.1093/mnras/stab1106}

\bibitem[{{Trampedach} {et~al.}(2014){Trampedach}, {Stein},
  {Christensen-Dalsgaard}, {Nordlund}, \& {Asplund}}]{Trampedach2014}
{Trampedach}, R., {Stein}, R.~F., {Christensen-Dalsgaard}, J., {Nordlund},
  {\r{A}}., \& {Asplund}, M. 2014, \mnras, 445, 4366,
  \dodoi{10.1093/mnras/stu2084}

\bibitem[{{van Saders} \& {Pinsonneault}(2012)}]{van2012}
{van Saders}, J.~L., \& {Pinsonneault}, M.~H. 2012, \apj, 751, 98,
  \dodoi{10.1088/0004-637X/751/2/98}

\bibitem[{Virtanen {et~al.}(2020)Virtanen, Gommers, Oliphant, Haberland, Reddy,
  Cournapeau, Burovski, Peterson, Weckesser, Bright, {van der Walt}, Brett,
  Wilson, Millman, Mayorov, Nelson, Jones, Kern, Larson, Carey, Polat, Feng,
  Moore, {VanderPlas}, Laxalde, Perktold, Cimrman, Henriksen, Quintero, Harris,
  Archibald, Ribeiro, Pedregosa, {van Mulbregt}, \& {SciPy 1.0
  Contributors}}]{2020SciPy-NMeth}
Virtanen, P., Gommers, R., Oliphant, T.~E., {et~al.} 2020, Nature Methods, 17,
  261, \dodoi{10.1038/s41592-019-0686-2}

\end{thebibliography}

\section{Updated Opacities}\label{sec:opac}
% Radiative opacity is fundamental to stellar structure, it determines how much
% incident radiation is absorbed or scattered. Moreover, when a media is in
% thermodynamic equilibrium with the radiation field, that is when the temperature
% of the media and that of the radiation field is the same, the opacity may be
% used via Kirchhoff's law to find the emissivity of a material
% \citep{Huebner2014}. Local Thermodynamic Equilibrium (LTE) is a common state to
% find within a star and therefore stellar models have long relied on opacities
% calculated in LTE.
Multiple groups have released high-temperature opacities including, the Opacity
Project \citep[OP][]{Seaton1994}, Laurence Livermore National Labs OPAL opacity
tables \citep{Iglesias1996}, and Los Alamos National Labs OPLIB opacity tables
\citep{Colgan2016}. OPAL high-temperature radiative opacity tables in
particular are very widely used by current generation isochrone grids
\citep[e.g. Dartmouth, MIST, \& StarEvol, ][]{Dotter2008,Choi2016,Amard2019}.
OPLIB opacity tables \citep{Colgan2016} are not widely used but include the
most up-to-date plasma modeling.

While the overall effect on the CMD of using OPLIB compared to OPAL tables is
small, the strong theoretical opacity dependence of the Jao Gap raises the
potential for these small effects to measurably shift the Gap's location. We
update DSEP to use high temperature opacity tables based on measurements from
Los Alamos national Labs T-1 group \citep[OPLIB,][]{Colgan2016}. The OPLIB
tables are created with ATOMIC \citep{Magee2004,Hakel2006,Fontes2016}, a modern
LTE and non-LTE opacity and plasma modeling code. These updated tables were
initially created in order to incorporate the most up to date plasma
physics at the time \citep{Bahcall2005}. 

OPLIB tables include monochromatic Rosseland mean opacities --- composed from
bound-bound, bound-free, free-free, and scattering opacities --- for elements
hydrogen through zinc over temperatures 0.5eV to 100 keV (5802 K -- 1.16$\times
10 ^{9}$ K) and for mass densities from approximately $10^{-8}$ g cm$^{-3}$ up
to approximately $10^{4}$ g cm$^{-3}$ (though the exact mass density range
varies as a function of temperature). 

DSEP ramps the \citet{Ferguson2005} low temperature opacities to high
temperature opacities tables between $10^{4.3}$ K and $10^{4.5}$ K; therefore,
only differences between high-temperature opacity sources above $10^{4.3}$ K
can effect model evolution. When comparing OPAL and OPLIB opacity tables
(Figure \ref{fig:opacComp}) we find OPLIB opacities are systematically lower
than OPAL opacities for temperatures above $10^{5}$ K. Between $10^{4.3}$ and
$10^{5} K$ OPLIB opacities are larger than OPAL opacities. These generally
lower opacities will decrease the radiative temperature gradient throughout
much of the radius of a model.

\begin{figure}
	\centering
	\includegraphics[width=0.45\textwidth]{OpacityComparision.pdf}
	\caption{Rosseland mean opacity with the GS98 solar composition for both
	OPAL opacities and OPLIB opacities (top). Residuals between OPLIB opacities
	and OPAL opacities (bottom). These opacities are plotted at $\log _{10}(R)
	= -0.5$, $X=0.7$, and $Z=0.02$. $\log _{10}(R)=-0.5$ approximates
	much of the interior a 0.35 M$_{\odot}$ model. Note how the OPLIB
	opacities are systematically lower than the OPAL opacities for temperatures
	above $10^{5.2}$ K.}
	\label{fig:opacComp}
\end{figure}

\subsection{Table Querying and Conversion}
The high-temperature opacity tables used by DSEP and most other stellar
evolution programs give Rosseland-mean opacity, $\kappa_{R}$, along three
dimensions: temperature, a density proxy $R$ (Equation \ref{eqn:R}; $T_{6} =
T\times10^{-6}$, $\rho$ is the mass density), and composition. 

\begin{align} \label{eqn:R}
	R = \frac{\rho}{T_{6}^{3}}
\end{align}

OPLIB tables may be queried from a web
interface\footnote{https://aphysics2.lanl.gov/apps/}; however, OPLIB opacities
are parametrized using mass-density and temperature instead of $R$ and
temperature. It is most efficient for us to convert these tables to the OPAL
format instead of modifying DSEP to use the OPLIB format directly. In order to
generate many tables easily and quickly we develop a web scraper
\citep[\texttt{pyTOPSScrape},][]{Boudreaux22} which can automatically retrieve
all the tables needed to build an opacity table in the OPAL format.
\texttt{pyTOPSScrape}\footnote{https://github.com/tboudreaux/pytopsscrape} has
been released under the permissive \texttt{MIT} license with the consent of the
Los Alamos T-1 group. For a detailed discussion of how the web scraper works
and how OPLIB tables are transformed into a format DSEP can use see Appendices
\ref{apx:pytopsscrape} \& \ref{apx:interp}.

\subsection{Solar Calibrated Stellar Models}\label{sec:SCSM}
% In order to further validate the OPLIB high-temperature opacities we first visually
% compare a set of opacity vs. temperature curves from OPLIB at a constant $R$
% and \citet{Grevesse1998} composition (GS98) to the same curve from OPAL. A
% characteristic opacity vs temperature curve is shown in Figure
% \ref{fig:OpacCompare}, $\log _{10}(R) = -1.5$ is chosen as for much of the
% radius of a main sequence star $\log _{10}(R)$ is around that value. The
% largest variation in $\kappa_{R}$ from OPAL to OPLIB at $\log _{10}(R)=-1.5$ is
% on the order of a few percent. This is inline with expectations of OPLIB and OPAL
% being in relatively close agreement \citep{Colgan2016}.


In order to validate the OPLIB opacities, we generate a solar calibrated
stellar model (SCSM) using these new tables. We first manually calibrate the
surface Z/X abundance to within one part in 100 of the solar value \citep[][Z/X=0.23]{Grevesse1998}.
Subsequently, we allow both the convective mixing length parameter,
$\alpha_{ML}$, and the initial Hydrogen mass fraction, $X$, to vary
simultaneously, minimizing the difference, to within one part in $10^{5}$,
between resultant models' final radius and luminosity to those of the sun.
Finally, we confirm that the model's surface Z/X abundance is still within one
part in 100 of the solar value.

\begin{figure}
	\centering
	\includegraphics[width=0.45\textwidth]{HRDiagramOPALvsOPLIB_SCCM.pdf}
	\caption{HR Diagram for the two SCSMs, OPAL and OPLIB. OPLIB is shown as a red
	dashed line.}
	\label{fig:OPLIBOPALHR}
\end{figure}

Solar calibrated stellar models evolved using GS98 OPAL and OPLIB opacity
tables (Figure \ref{fig:OPLIBOPALHR}) differ $\sim 0.5\%$ in the SCSM hydrogen
mass fractions and $\sim 1.5\%$ in the SCSM convective mixing length parameters
(Table \ref{tab:SCSMResults}). While the two evolutionary tracks are very
similar, note that the OPLIB SCSM's luminosity is systematically lower past the
solar age. While at the solar age the OPLIB SCSM luminosity is effectively the
same as the OPAL SCSM. This luminosity difference between OPAL and OPLIB based
models is not inconsistent with expectations given the more shallow radiative
temperature gradient resulting from the lower OPLIB opacities

\begin{table}
	\centering
	\begin{tabular}{l c c}
		\hline
		Model & $X$ & $\alpha_{ML}$ \\
		\hline
		\hline
		OPAL & 0.7066 & 1.9333 \\
		OPLIB & 0.7107 & 1.9629
	\end{tabular}
	\caption{Optimized parameters for SCSMs evolved using OPAL and OPLIB high
	temperature opacity tables.}
	\label{tab:SCSMResults}
\end{table}

